CS 252 Third Project Questionnaire Due Monday, April 13 by 5 PM Mail questionnaire or URL to gebis@eecs, or place in box in 283 Soda

Project Name:

Partner #1: email: Partner #2: email: Partner #3: email:

What work have you done on the project since the last project meeting?

Do you have any preliminary results?

What remains to be completed for the project?

What relevant literature have you found?

Are there any computing resources, drive space, or similar problems?

Any other comments or questions?

Please update your web page to reflect any results, literature, or project changes.